Author's response to reviews

Title: Audit of therapeutic interventions in inpatient children using two scores: are they evidence-based in developing countries?

Authors:

Nilton Y Carreazo (yhuro@hotmail.com)
Carlos A Bada (ifocar@yahoo.com)
Juan P Chalco (jpcho33@yahoo.com)
Luis Huicho (lhuicho@viabcp.com)

Version: 9 Date: 27 December 2004

Author's response to reviews:

Lima, December 27, 2004

Dr. Alison Z Shaw, Ph.D.
Assistant Editor
editorial@biomedcentral.com

Dear Dr. Shaw,

I found some minor errors in the last version of our submitted manuscript and made the corresponding modifications. These are the changes made:

Page 10, second paragraph, line 19:
"75 mml/l" was replaced by "75 mmol/l"

"Cellulites" was replaced by "cellulitis" in Tables 2 and 6.

"Hemophilia" was replaced by "Haemophilia" in Tables 2, 4 and 5.

"vase-occlusive" was replaced by "vaso-occlusive" in Tables 2, 4 and 5.

"PSVT" meaning was missing as a footnote in Tables 4 and 5. It is a shorthand for paroxysmal supraventricular tachycardia. It is included now as a footnote in such tables.

I hope the manuscript is ready now for publication in BMC.

Yours sincerely,

Luis Huicho, MD